Author’s response to reviews

Title: Improved Student Learning Outcomes Using a Blended Learning Approach in a Graduate-Level Public Health Course

Authors:

Marc T Kiviniemi (mtk8@buffalo.edu)

Version: 2 Date: 23 October 2013

Author’s response to reviews:

Both requested changes were made -- a statement about human subjects review was added at the beginning of the methods section. A statement of acknowledgements was added following the competing interests statement.